Author's response to reviews

Title: Small-bowel mucosal changes and antibody responses after low- and moderate-dose gluten challenge in celiac disease

Authors:

Marja-Leena Lähdeaho (marja-leena.lahdeaho@uta.fi)
Markku Mäki (markku.maki@uta.fi)
Kaija Laurila (kaija.laurila@uta.fi)
Heini Huhtala (heini.huhtala@uta.fi)
Katri Kaukinen (katri.kaukinen@uta.fi)

Version: 5 Date: 23 November 2011

Author's response to reviews:

Dear Editor-in-Chief,

thank you for the proof of our manuscript; Small bowel mucosal and antibody responses after low- and moderate gluten challenge in celiac disease (MS: 1459477404583655).

Two corrections are made to this version: page 10, line 11 the days are in right order. On page 12, line 7 the right figure number is 5. The revised sentences are highlighted in red.

Sincerely yours,
Katri Kaukinen, corresponding author